% This file is part of Oaklisp.
%
% This program is free software; you can redistribute it and/or modify
% it under the terms of the GNU General Public License as published by
% the Free Software Foundation; either version 2 of the License, or
% (at your option) any later version.
%
% This program is distributed in the hope that it will be useful,
% but WITHOUT ANY WARRANTY; without even the implied warranty of
% MERCHANTABILITY or FITNESS FOR A PARTICULAR PURPOSE.  See the
% GNU General Public License for more details.
%
% The GNU GPL is available at http://www.gnu.org/licenses/gpl.html
% or from the Free Software Foundation, 59 Temple Place - Suite 330,
% Boston, MA 02111-1307, USA

\documentclass[12pt]{report} % Blake McBride suggests [...,twoside]{book}
\usepackage{times}
\usepackage{fullpage}
\usepackage{graphicx}
\usepackage{makeidx}
\usepackage[numbers]{natbib}
\usepackage[hyphens]{url}
\urlstyle{same}

\makeindex

\begin{document}

\input{../mandefs}

\include{cover}
\tableofcontents
\newpage
\pagenumbering{arabic}
\include{intro}
\include{types}
\include{methods}
\include{sides}
\include{locales}
\include{dynamic}
\include{control}
\include{sequences}
\include{numbers}
\include{io}
\include{misc}
\include{user}

\nocite{OAKLANG88}
\nocite{CLOOPS}
\nocite{FLAVORS2}
\nocite{MVC}
\nocite{R3RS}
\nocite{SCHEME-DECLARATIVE}
\nocite{SCHEME-PAP}
\nocite{SNYDER86}
\nocite{T}
\nocite{T-MAN}
\nocite{OAK-PAP}
\nocite{3LISP}
\nocite{FLAVORS1}
\nocite{CLtL}
\nocite{MULTILISP85}
\nocite{ACTORS78}
\nocite{CITY-TRASH}
\nocite{PEARLMUTTER-LANG90A}
\nocite{PEARLMUTTER99}

\bibliography{../oakman}

\printindex

\end{document}
